Author’s response to reviews

Title: Trial and error: Evaluating and refining a community model of HIV testing in Australia

Authors:

Kathleen Ryan (kathleen.ryan@burnet.edu.au)
Alisa Pedrana (alisa.pedrana@burnet.edu.au)
David Leitinger (david.leitinger@pronto.org.au)
Anna Wilkinson (anna.wilkinson@burnet.edu.au)
Peter Locke (peter.locke@vac.org.au)
Margaret Hellard (margaret.hellard@burnet.edu.au)
Mark Stoove (mark.stoove@burnet.edu.au)

Version: 3 Date: 21 Sep 2017

Author’s response to reviews:

1. Please change the "Discussion" heading in the abstract to "Conclusions".

This subheading has been updated

2. Please add a “Conclusions” section after the “Discussion” section in the main text. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

A conclusion subheading has been added

3. Please include an "Abbreviations" section to the Declarations

An abbreviations section has been added